Author’s response to reviews

Title: A real-life observational study of the effectiveness of FACT in a Dutch mental health region

Authors:

Marjan Drukker (Marjan.drukker@sp.unimaas.nl)
Myrte Maarschalkerweerd (MI.Maarschalkerweerd@alumni.unimaas.nl)
Maarten Bak (M.Bak@sp.unimaas.nl)
Ger Driessen (G.Driessen@sp.unimaas.nl)
Joost À Campo (J.a.Campo@smzg.nl)
Arthur De Bie (A.deBie@prinsclauscentrum.nl)
Giovanni Poddighe (G.Poddighe@riagg-maastricht.nl)
Jim Van Os (J.vanOs@sp.unimaas.nl)
Philippe Delespaul (Ph.Delespaul@sp.unimaas.nl)

Version: 4 Date: 28 November 2008

Author’s response to reviews:

I deleted all vertical lines in the tables
I deleted the merged cell across rows with the vertical text and added a horizontal text instead.
I deleted the hyperlinks from the e-mail adresses

I changed 'List of abbreviations' to 'Abbreviations' and I deleted the hard returns in this section and replaced them with within-paragraph returns. I hope this is meant by "format as a paragraph rather than a list."

I added the heading 'Tables' to the tables section.